

# WORKING PAPER SERIES NO. 313 / FEBRUARY 2004

THE HIGH-YIELD SEGMENT OF THE CORPORATE BOND MARKET: A DIFFUSION MODELLING APPROACH FOR THE UNITED STATES, THE UNITED KINGDOM AND THE EURO AREA

by Gabe de Bondt and David Marqués



# WORKING PAPER SERIES

NO. 313 / FEBRUARY 2004

THE HIGH-YIELD SEGMENT OF THE CORPORATE BOND MARKET: A DIFFUSION MODELLING APPROACH FOR THE UNITED STATES, THE UNITED KINGDOM AND THE EURO AREA'

> by Gabe de Bondt<sup>2</sup> and David Marqués<sup>2</sup>

This paper can be downloaded without charge from http://www.ecb.int or from the Social Science Research Network electronic library at http://ssrn.com/abstract\_id=515076.





a motif taken from the €100 banknote.



 We are grateful for comments by Edward Altman, Jesper Berg, Hans-Joachim Klöckers, Andrés Manzanares, Phil Molyneux, several Merrill Lynch Staff Members from the Global Securities Research & Economics Group, and an anonymous referee. All views expressed are those of the authors alone and do not necessarily reflect those of the ECB or the Eurosystem.
 2 European Central Bank, Kaiserstrasse 29, D-60311, Frankfurt am Main, GERMANY.Tel.:+49 69 1344 6477/6460; fax: +49 69 1344 6514; e-mail address: gabe.de\_bondt@ecb.int and david.marques@ecb.int.

### © European Central Bank, 2004

Address Kaiserstrasse 29 60311 Frankfurt am Main, Germany

Postal address Postfach 16 03 19 60066 Frankfurt am Main, Germany

**Telephone** +49 69 1344 0

Internet http://www.ecb.int

Fax +49 69 1344 6000

**Telex** 411 144 ecb d

All rights reserved.

Reproduction for educational and noncommercial purposes is permitted provided that the source is acknowledged.

The views expressed in this paper do not necessarily reflect those of the European Central Bank.

The statement of purpose for the ECB Working Paper Series is available from the ECB website, http://www.ecb.int.

ISSN 1561-0810 (print) ISSN 1725-2806 (online)

# CONTENTS

Ab	ostract	4					
Non-technical summary							
1	Introduction						
2	High-yield bond market: key characteristics	9					
3	Previous studies on the diffusion of financial innovations	13					
4	High-yield bond diffusion model specifications	14					
	4.1 Standard diffusion models	14					
	4.2 Diffusion models with	15					
5	Data	17					
6	Empirical results	21					
0	6.1 Empirical diffusion models	21					
	6.2 Estimates	22					
	6.3 Comparison of high-yield bond diffusion rates	26					
7	Concluding remarks	28					
Annexes							
Re	ferences	33					
Eu	European Central Bank						
working paper series 36							

S

### Abstract

This study empirically examines the development of the high-yield segment of the corporate bond market in the United States, as a pioneer country, and the United Kingdom and the euro area, as later adopting countries. Estimated diffusion models show for the United States a significant pioneer influence factor and autonomous speed of diffusion. The latter is found to be higher in Europe than in the United States as also macroeconomic factors are considered. The high-yield bond diffusion pattern is significantly affected by financing need variables, e.g. leverage buy-outs, mergers and acquisitions, and industrial production growth, and return or financing cost variables, e.g. stock market return and the spread between the yield on speculative-grade and BBB-rated investment-grade bonds. These findings suggest that the diffusion of new financial products depends on the macroeconomic environment and can be quickly in case of the diffusion from a pioneer country to later adopting countries.

Keywords: high-yield bond market; financial innovation; diffusion models

JEL classification: G32; E44

### Non-technical summary

An important feature of a well-developed financial system is the existence of a robust corporate bond market working alongside a sound banking system. The existence of a mature corporate bond market including a deep high-yield bond segment, i.e. below investment-grade rated corporate bonds, appears to be positive for economic development. It allows corporations to raise funds more quickly and on more flexible terms than would otherwise have been available. The existence of this avenue of corporate finance complementary to bank and equity finance is particularly beneficial for economies with a large number of small and medium sized firms.

This study aims to provide empirical insights into the development of the high-yield segment of the corporate bond market in the United States, as a pioneer or early adopting country, and the United Kingdom and the euro area, as imitating or later adopting countries. In order to model the pattern of growth of the high-yield bond market, high-yield bonds are viewed as a financial innovation. High-yield bond issuance is expected to grow over time as the high-yield bond financial innovation is diffused among corporations. As such, it is natural to model its growth pattern by applying the diffusion models utilised in the literature for the adoption of new products or technologies, such as the spread of medical diseases or the adoption of new telecommunication technologies.

Along these lines of applying a diffusion modelling approach, the focus of this study is on the relative size of the high-yield bond market: the market share of high-yield bonds in the overall corporate bond market in terms of amounts outstanding. Starting from a very low base, the market share of high-yield bonds in the United States grew very fast during the 1980s, but settled down to around 6% in the early 1990s, a share it has oscillated around ever since. During the 1980s, the United States market for junk bonds was set in motion to facilitate the financing of highly leveraged or start-up companies, which were unable to raise funds from banks or were unable to borrow on flexible conditions. High-yield bond issuance peaked in the late 1980s, subsequently dropping off in the early 1990s as the United States went through a recession and heightened credit risk concerns. The high-yield bond market in Europe, divided into the United Kingdom and the euro area, which had shown some incipient signs of developing as early as mid-1997, began to grow substantially in the run up of the introduction of the euro. The period at which high-yield bonds started to be issued up to the peak in the high-yield segment of the corporate bond market, the so-called diffusion sample period, covers for the United States January 1981 to August 1988, for the United Kingdom December 1998 to July 2000, and for the euro area December 1998 to January 2001. For the United Kingdom and the euro area also estimation results on the basis of a full sample up to December 2001 are shown, because we can not yet be sure whether the observed peaks in the high-yield bond market penetration are local or absolute maxima.

Three key potential determinants of the high-yield bond diffusion pattern are considered. The first determinant is a pioneer influence factor. This explanatory variable reflects the role of innovators and affects the years needed to reach a peak in the cumulative adoption of high-yield bonds. The second explanatory factor is an autonomous speed of diffusion. The larger this rate of imitation, the more rapid

the diffusion of the financial innovation. The third set of determinants is macroeconomic factors, since the development of the high-yield bond market may depend on the macroeconomic conditions. Regression results of different diffusion models indeed show that these three determinants play a significant role in the diffusion process of high-yield bonds. In the United States a pioneer influence factor and an autonomous speed of diffusion predominantly determine the diffusion of high-yield bonds. In the United Kingdom and the euro area, not surprisingly, the former is not found to be relevant, while the latter is found to be faster than in the United States, at least as macroeconomic factors are taken into account. The high-yield bond diffusion is found to be significantly affected by both the corporate financing needs, e.g. leverage buy-outs, mergers and acquisitions, and the industrial production growth, and return and financing cost considerations, e.g. stock market return and the spread between the yield on speculative-grade and BBB-rated investment-grade bonds.

The above suggests two main messages. The first is our main contention that high-yield bonds are diffused in the United States and other countries as a financial innovation, whereby the macroeconomic environment, fully neglected in previous studies, plays a significant role. The second, more tentative in nature but supported by our empirical findings, is that the high-yield bond diffusion in the United States has been substantially slower than in European countries. The United States was an early adopter where the high-yield bond technology became gradually available, while for the European countries, as late adopters, all information concerning the financial innovation was immediately available at the beginning of the diffusion period. Consequently, the financial landscape in the later adopting countries might change within a short time span because of a quick adoption of already existing "foreign" financial products from a pioneer country, e.g. the United States, towards later adopting countries, e.g. the United Kingdom and euro area countries.

From a modelling point of view, this study provides a framework, which takes account of the macroeconomic environment and may form the starting point for future research on the diffusion of other financial innovative products, for instance all type of securitised assets. Another promising avenue for future research is the diffusion of financial innovations across regions, for instance from industrialised countries to emerging market economies.

### 1. Introduction

In recent years, a growing branch of the economic literature has focused on the relationship between financial structure and the real economy. An important finding of theoretical and empirical studies is the existence of a causal link between financial development and economic growth.<sup>1</sup> The main reasons for this link are twofold. First, a developed financial system allocates funds into their most profitable uses. Second, it stabilises consumption by allowing individuals to diversify their investments more effectively. An important feature of a well-developed financial system is the existence of a robust corporate bond market working alongside a sound banking system (Herring and Chatusripitak, 2001, de Bondt, 2002). Within the corporate bond market, high-yield bonds form a particularly interesting segment for at least four main reasons.

First, high-yield bonds provide, under usual credit conditions, a larger degree of flexibility than bank loans which are subject to more strict covenants and narrower investment conditions. These flexibility features of high-yield bonds are particularly enticing for fast-growing firms that tend to rely extensively on bank financing (Gilson and Warner, 1997). More generally, a well-developed high-yield bond market could be beneficial towards smoothening the financing of firms as the junk bond market could provide funds complementary to bank-based debt or equity (Allen and Gale, 2000, and Davis 2001).

Secondly, financing via high-yield bonds can encourage a swifter reallocation of funds from cash-rich but economically declining sectors, to fast-growing sectors with urgent needs of funds. Consequently, a well-developed financial sector, in which there is a deep and liquid market for bonds which are either rated below investment grade or unrated, should facilitate both innovative new business and the transition of medium sized firms into large enterprises (Rajan and Zingales, 1998). Corporations which issue high-yield bonds have typically outperformed industrial averages in terms of industrial performance, such as employment growth, productivity and capital investment (Yago and Trimbath, 2003).

Thirdly, the market pricing of speculative-grade bonds takes place on a continuous basis by the interplay of market participants. In this way, the quality of credit is monitored by a large number of market participants on a continuous basis. Consequently, a financial system with a well-developed high-yield bond market provides immediate discipline for lower credit quality corporations' activities via prices of their marketable corporate liabilities. In this sense, by continuously monitoring firms performance, the creation of this segment of the corporate bond market also favours the creation of value-added by aligning the incentives of managers with those of shareholders (Altman and Smith, 1991).

Finally, from a macroeconomic perspective the high-yield segment of the corporate bond market could be a useful source of information on future economic activity and on current credit conditions in the economy (Gertler and Lown, 1999, Cooper et al., 2001, and Zhang, 2002). Price spreads over other financial instruments as well as issuance conditions on the high-yield bond market may capture the general degree of concern in the economy about credit risk. More generally, the development of the high-

<sup>&</sup>lt;sup>1</sup> Although this literature has its roots in the 1960s and early 1970s (Goldsmith, 1969), it has recovered prominence over the last decade (Stulz, 2000). For a more specific review focusing on the effect of the corporate bond market on economic development, see Herring and Chatusripitak (2001).

yield bond market may affect the costs and scope of financial intermediation and therefore the process of monetary policy transmission.

Against this background, this study aims to provide empirical insights into the development of the highyield segment of the corporate bond market. A diffusion modelling approach is adopted by considering high-yield bonds issued by residents and denominated in the national currency as a financial innovation. This approach seems natural because the high-yield bond market activity is expected to grow over time as the high-yield bond financial innovation is diffused among corporations. The model also takes account of macroeconomic factors that may have an impact on the diffusion process of high-yield bonds. Furthermore, the diffusion adoption pattern in the United States in the 1980s, which represents the time in which junk bonds started to diffuse, is compared with the prevailing one in Europe, divided into the United Kingdom and the euro area, which had shown some incipient signs of developing as early as mid-1997. A more general purpose of this study is to provide a modelling framework which can be used as a starting point for the examination of the diffusion of other financial innovations and across countries.

The main message of this study is twofold. The first is our main contention that high-yield bonds are diffused as a financial innovation, whereby the macroeconomic environment plays a significant role. The second is our suspicion, but supported by our empirical results, that the high-yield bond diffusion in the United States has been substantially slower than in Europe. The United States was an early adopter where the high-yield bond technology became gradually available, while for the later adopting European countries all information concerning the financial innovation was immediately available at the beginning of the diffusion period. Consequently, the financial landscape in later adopting countries might change within a short time span because of a quick diffusion of financial products from a pioneer or early adopting country, e.g. the United States, towards imitating or later adopting countries, e.g. the United Kingdom and euro area countries.

From an empirical modelling point of view, two main conclusions emerge. First, the Bass model, which besides a rate of imitation or autonomous speed of diffusion also models a pioneer influence factor or rate of innovation, seems to be typically the most statistically appropriate diffusion model for the United States. The popular Mansfield logistic diffusion model, which only models an autonomous speed of diffusion, is found to be the most appropriate diffusion model for the high-yield bond adoption in Europe. The second empirical conclusion is that besides a rate of innovation and diffusion, macroeconomic factors are relevant in the determination of the high-yield bond diffusion. Diffusion models including macroeconomic factors show that the high-yield bond diffusion slows down following a rise in M&A activity. At the same time, it is found that leverage buy-outs and stock market developments positively affect the adoption of high-yield bonds and that the impact of business cycle conditions is ambiguous. On the one hand, the financing needs of high-yield bond issuers, for instance high-growth companies, tend to be relatively cyclical. On the other hand, during economic downturns the share of "fallen angels" in the high-yield segment of the corporate bond market, that is the downgrading of investment-grade bonds to junk quality, rises, implying a negative relationship between high-yield bond diffusion and industrial production. The latter effect is found to dominate during the core diffusion sample period for the United

States and the former for the United Kingdom and the euro area. The spread between the yield on speculative-grade and BBB-rated, the most risky investment-grade, bonds is found to be insignificant for the high-yield bond diffusion process in the United States and the euro area. In the United Kingdom, however, this spread has significantly positively affected the high-yield bond diffusion process, suggesting that investor return considerations (demand factor) has dominated financing cost thoughts (supply factor).

The remainder of this study is organised as follows. Section 2 provides an introductory note on the key characteristics of the high-yield bond market. Section 3 highlights the literature on the diffusion of financial innovations. Section 4 introduces high-yield bond diffusion models. Section 5 describes the data. Section 6 presents and discusses the empirical results based on diffusion models without and with macroeconomic variables. Section 7 provides concluding remarks. Finally, Annexes 1, 2 and 3 describe the data in detail, plot diffusion patterns based on different values for the estimated diffusion model parameters, and provide estimation results for different market penetration levels, respectively.

## 2. High-yield bond market: key characteristics

Although the high-yield corporate bond market is normally described as a single market, it encompasses a wide range of different high-yield instruments: straight non-convertible bonds, exchangeable variable rate notes, increasing rate notes, reset notes, extendable notes, commodity linked bonds, usable bonds and springing issues, among others. These bonds can be viewed as substitutes for bank loans and private placements, which the original lenders typically hold until maturity. In other words, high-yield bonds can be thought of as term loans that have been packaged to enhance their liquidity and divisibility.

The creation in the United States of the high-yield segment of the corporate bond market in the late 1970s and early 1980s is viewed as a major financial innovation of the fixed income market over the last decades. Prior to the early 1980s, the high-yield bond market in the United States consisted entirely of bonds issued by "fallen angels" (Taggart, 1988). During the 1980s, the United States market for junk bonds was set in motion to facilitate the financing of highly leveraged or start-up companies, which were unable to borrow on flexible conditions or to raise funds from banks. High-yield bond issuance peaked in the late 1980s, subsequently dropping off in the early 1990s as a rise in credit risk concerns was observed, as triggered by a massive increase in corporate defaults in the United States, a business recession and a poorly performing stock market. As defaults on high-yield bonds mounted, the market's leading underwriter and trader of high-yield debt securities, Drexel Burnham Lambert, went bankrupt and several major investors were barred from buying new high-yield bonds and were forced to liquidate their high-yield bond positions (Ma et al., 1989, Altman, 1992, Brewer and Jackson, 2000). Due to its role on the high-yield market, the collapse of Drexel Burnham Lambert had a major impact on the market in terms of issues and prices. After its collapse the market for newly issued high-yield bonds almost ground to a halt and no significant new issue came to the market for more than a year (Brewer and Jackson, 2000).

Later on, after 1993, as interest rates declined and the economy recovered, many issuers returned to the market to refinance old debt at the lower interest rate levels prevailing at that time. New issuers also came to the market during the second half of the 1990s. Despite a record number of bond defaults in 2001 and a flight to quality, following the terrorist attacks in September 2001, the high-yield bond market displayed impressive resiliency (Altman and Arman, 2002). Nowadays the market has broadened to include a variety of issuers with diverse financing needs, such as funding for working capital for growing companies.

While the US high-yield bond market has flourished over the last two decades, in other countries this market has only begun to develop in the late 1990s and remains both in terms of amounts outstanding and gross issuance well below the US market (see Charts 1 and 2). High-yield bonds have remained an overwhelmingly US development with little exposure to the global markets except initially for a second order effect, being primarily the operational exposure to the international markets from US-based corporations, e.g. American Standard's European and Far East operations, etc. (DeRosa-Farg and Blau, 1999). Besides the United States this study focuses on Europe, the second largest region of the global high-yield bond market. In this respect, Europe is divided into the United Kingdom and the euro area. Other regions with some high-yield bond issuance activity are Canada, Latin America and Asia. High-yield bonds in Canada are predominantly denominated in the US dollar instead of the Canadian dollar (Miville and Bernier, 1999). This study, however, examines high-yield bonds issued by residents denominated in the national currency. In Japan the abolition of issue standards in January 1996 has made possible the issuance of high-yield bonds, enabling a, still rather low, number of companies to engage in this type of financing.

Turning to Europe, in the early 1990s some authors were already arguing that the development of this market in Europe was likely to take place in the wake of the introduction of the Single Market for Financial Services in 1993 (Molyneux, 1990). It was argued that from a demand perspective, European institutional investors had been actively investing in the US junk bond market. Also from a supply side, management buy-outs had for some time been financed in Europe with mezzanine debt finance.<sup>2</sup> However, despite these factors, the European high-yield bond market, which had shown some incipient signs of developing as early as mid-1997,<sup>3</sup> began only to grow substantially around the start of Stage Three of EMU. This development seems to be related to expectations of a truly integrated financial market, which in turn was triggered by the launch of the euro (Marqués et al., 2000, ECB, 2000 and 2001). Besides the catalytic impact of the euro, two major underlying factors seem to spear the development of the high-yield bond market in the euro area (Altman, 1998). First, the euro area financial landscape encompassed a large number of small and medium-sized companies in need of finance but lacking the size and earnings predictability to obtain investment-grade ratings. Second, bond financing provides an instrument to restructure firms that are financed with usually more restrictive bank loans.

<sup>&</sup>lt;sup>2</sup> Mezzanine debt is an European term used to describe the unsecured component of a management or leverage buy-outs and shares a number of features with high-yield bonds (Molyneux, 1990). Mezzanine debt tends to be sold to a smaller investor group and therefore is less rigid and quicker to arrange than high-yield bonds.

<sup>&</sup>lt;sup>3</sup> Figures are from Bondware. High-yield bond issued by residents and denominated in currencies other than the national one, took, however, incidentally place in the 1990s.

This is particularly the case in the euro area due to the predominant role that banks play in most euro area countries.



Turning to the currency of issuance in Europe, the proportion of European borrowers issuing high-yield bonds denominated in euro has been systematically increasing over the last years at the expense of dollar-denominated issues (see Table 1). Despite this advancement, the role of the euro in the high-yield global market remains limited (see Chart 2). In terms of ratings, the proportion of BB rated bonds as a percentage of total high yields issuance has been increasing since the inception of the high-yield bond market in Europe, suggesting an increase on perceived credit quality among high-yield European issuers (see Table 2).

	Currency				Total volume
Year	Euro (%)	Sterling (%)	US Dollar (%)	Other (%)	(\$ millions )
1998	8.4	12.8	66.2	12.5	13,321
1999	36.3	10.4	53.4	5.4	17,834
2000	56.9	13.2	29.0	5.3	14,058
2001	74.8	17.0	8.1	5.0	6,758

 Table 1 European high-yield new issuance by currency distribution

Source: Merrill Lynch.

Year	Volume (\$ millions)	BB	В	CCC/CC/C	Not rated
1998	13,321	3%	81.5%	9.1%	5.5%
1999	17,834	12.3%	75.3%	5.4%	7.0%
2000	14,058	12.1%	80.6%	5.3%	2.0%
2001	6,758	30.1%	63.1%	5.0%	1.7%



Sources: Merrill Lynch and Moody's.

Concerning the sector composition, the telecommunication, media and technology (TMT) and consumer sector hold a more important weight at the euro-denominated high-yield bond market than at the more consolidated US dollar-denominated market. The latter shows a sector composition which is more heterogeneous and less dependent on a limited number of predominant issuers and sectors (see Charts 3 and 4).



Finally, it is noteworthy that the bulk of euro-denominated high-yield bonds has US banks as underwriters (see Table 3). This may be explained by the fact that US banks have considerable experience in pricing credit risk. Euro area banks more or less only started to focus on credit risk in the run up of the introduction of the euro. With the advent of the euro, credit risk has become more imperative in pricing securities than the knowledge of national factors (read: exchange rate risk). The main possible source of comparative advantage for European investment banks in terms of high-yield bond underwriting, namely a closer relationship with the customer and a better understanding of the national accounting, legal, fiscal and monetary environment have been eroded, owing to the introduction of the euro. Meanwhile, US banks have been generally quicker to develop an analytical and marketing capacity at a pan-European level.

		<b>. . . .</b>
Position	Bookrunner	Share in % of total
1	Goldman Sachs	16
2	Merrill Lynch	14
3	Credit Suisse First Boston	14
4	Salomon Brothers	13
5	Lehman Brothers	5
6	Barclays Capital	5
7	WestLB Panmure	5
8	ABN AMRO Rothschild	5
9	Morgan Stanley	3
10	Banca Commerciale Italiana	2
<i>a b</i>		

 Table 3 Underwriters of euro-denominated high-yield bonds, 1998-2001

Source: Bondware.

### 3. Previous studies on the diffusion of financial innovations

The finance literature provides a wide range of frameworks for the analysis of financial innovations. It investigates whether and how a financial innovation is affected by a) external environmental factors (which could be of an economic, regulatory and/or technological nature), b) the interplay between supply and demand forces on the financial innovation process, or c) product characteristics (Allen and Gale, 1994, and Jesswein et al., 1996). For example, Tufano (1989) examines for the United States 58 financial innovations during the years 1974–1986, including mortgage-backed securities, asset-backed securities, non-equity-linked bonds and equity-linked bonds. By using a simple profit maximising framework, he analyses how investment bankers are compensated for their investments in developing new financial products. The empirical evidence on prices and quantities suggests that banks do not charge higher prices for innovations than with imitative products. Table 3 in the previous section suggests a similar phenomenon for high-yield bonds, US banks have captured as innovators a large market share in the euro-denominated high-yield bond market.

Another strand of the literature focuses on the empirical functional form of the diffusion process of new products. Starting from the seminal work on agricultural economics by Griliches (1957), this kind of diffusion model has been widely applied to different problems, ranging from models dealing with the spread of medical diseases (Davies, 1979) to the diffusion of new telecommunication technologies (Gruber and Verboven, 2001). In the field of financial innovation, several studies have followed a diffusion modelling approach to model the adoption rate of new financial products. Jagtiani et al. (1995) and de Bondt (2000) examine off-balance sheet activities in the United States and Europe, respectively, while Molyneux and Shamroukh (1996) analyse the high-yield bond diffusion process in the United States.

As regards the off-balance sheet diffusion, the substantial growth in off-balance sheet activities can be viewed as a reflection of the growing importance of securitisation. Jagtiani et al. (1995) examine for the United States the impact of bank capital regulation on the adoption pattern of five off-balance sheet products, e.g. standby letters of credit, loan sales, swaps, options, and futures and forwards. Their results, based on the logistic diffusion modelling approach of Mansfield (1961), suggest that neither changes in

capital regulation nor bank-specific characteristics have had a consistent effect on the adoption of offbalance sheet products. De Bondt (2000) extends the work of Jagtiani et al. (1995) by also considering macroeconomic conditions as determinants of OBS diffusion. His regression results show that an autonomous speed of diffusion, bank-specific characteristics, regulatory and macroeconomic factors have significantly affected the off-balance sheet diffusion in Europe during 1989–1995.

Turning to the diffusion process of high-yield bonds, Molyneux and Shamroukh (1996) examine this issue within the Mansfield (1961) model and two other diffusion models for the United States during the years 1977–1986.<sup>4</sup> They decompose the number of adopters into two categories, namely external adopters or innovators and internal adopters or imitators. External adopters are influenced in their adoption by the initial exogenous factors, such as changes in the regulatory environment or demand factors. Internal adopters are, in turn, affected by bandwagon pressures of institutional and/or competitive nature and also by their updated cost/benefit assessment of adopting the innovation due to endogenous factors, the so-called rational efficiency effects. Their results suggest that both external and internal adoption factors were significant in the diffusion pattern of high-yield bonds in the United States.

Overall, a selected review on previous studies reveals that there is a paucity of empirical work on the diffusion of financial innovations. Also, from a research perspective, when trying to model diffusion processes of financial innovation, it is important to focus both on the empirical functional form of the diffusion process as well as on external economic determinants, which are likely to affect the diffusion process.

# 4. High-yield bond diffusion model specifications

From the previous sections, it is clear that in order to analyse the pattern of growth of the high-yield segment of the corporate bond market, three main factors have to be taken into account. First, the high-yield bond market is a financial innovation, and as such, it is natural to model its pattern of growth by applying the diffusion models utilised in the financial innovation literature. Second, the development of this market seems to be dependent on a number of macroeconomic factors, which are likely to influence the progress of this market in conjunction with the financial innovation dynamics. Finally, and due to the earlier development of the market in the United States, there is likely to be a pioneer influence effect for the United States, but not necessarily for later adopting countries.

# 4.1 Standard diffusion models

This section introduces standard new product diffusion models as reviewed by Mahajan et al. (1990), Parker (1994) and Geroski (1999). These diffusion model generally generate an S-shape or sigmoidal pattern for the cumulative adoption over time and a bell-shaped diffusion curve for new adopters.

<sup>&</sup>lt;sup>4</sup> They also examine the diffusion pattern of note issuance facilities in the United States during 1983–1986.

Due to its simplicity, the majority of studies use the Mansfield (1961) model, which was first applied in economics by Griliches (1957)<sup>5</sup> to model the diffusion pattern of hybrid corn in the United States. The Mansfield model has been successfully empirically applied in case of a high diffusion rate or word-of-mouth effect. Some of the restrictions of this model were successfully relaxed by the Bass (1969) model, which postulates the following diffusion model:

$$\Delta N(t) = n(t) = \left[a + b\left(\frac{N(t)}{M}\right)\right] \cdot \left[M - N(t)\right]$$
<sup>[1]</sup>

With M defined as the *market potential* or more graphically as the peak of the ultimate number of total adopters,<sup>6</sup> n(t) as the number of adopters in time t, and N(t) as the total number of cumulative adopters at time t. In other words,  $\Delta N(t)$  equals n(t). Coefficient a represents the initial proportion of innovators or pioneer influence factor, and coefficient b is the proportion of imitators or the rate of diffusion or more intuitively the speed of adoption. The main advantage of this model is that it generalises the earlier models by Fourt and Woodlock (1960), which sets b=0, and Mansfield (1961), which assumes a=0.

During the late 1970s and 1980s the original Bass model was extended in various ways.<sup>7</sup> Foremost among the proposed model's extensions were a dynamic market potential by making the market potential dependent on time (M(t)) (Mahajan and Peterson, 1978). Other improvements were the possibility to consider a long-term term *market penetration ceiling*, defined by the coefficient *c*, that allows for non-adoption for a percentage of the market potential. Also the possibility of having a non-symmetric diffusion curve or *non-uniform influences* represented by the coefficient *d* (Easingwood et al., 1983) or to allow for *heterogeneity in the adopter population*, indicated by the coefficient *e* (Jeuland, 1981). These modifications of the Bass model result in the following five-parameter innovation diffusion model.<sup>8</sup>

$$n(t) = \left[a + b\left(\frac{N(t)}{cM(t)}\right)^{d}\right] \cdot \left[cM(t) - N(t)\right]^{e}$$
[2]

### 4.2 Diffusion models with macroeconomic variables

For our purpose, a caveat of standard diffusion models is that they do not combine the "contagion or diffusion effects" with other factors that may play a role in the determination of high-yield bond diffusion (Bass et al., 1994). The diffusion process is made dependent on a range of macroeconomic variables likely to influence it over time by adding macroeconomic variables linearly to the main diffusion equation. The effects of the macroeconomic explanatory variables are influenced by the diffusion process

<sup>&</sup>lt;sup>5</sup> These functions had already been applied in other fields, see for instance Winsor (1932).

<sup>&</sup>lt;sup>6</sup> Economically the upper limit represents the achievement of the market potential. The mathematical derivation of the logistic curve used by Griliches (1957) and Mansfield (1961) is a bounded continuos time function in which the lower limit is 0 and the upper limit is a bounded upper limit, the market potential or peak.

<sup>&</sup>lt;sup>7</sup> In addition to Bass-type diffusion models, any number of growth models can be used to characterise cumulative distribution functions, and thus S-shaped diffusion curves. Early examples are for instance the Gompertz curve or the Floyd model.

<sup>&</sup>lt;sup>8</sup> It is worth noting that within equation [2] there are some 20-nested models which have all different characteristics and seen some empirical application in the diffusion literature.

via the multiplicative inclusion of a dampening effect with the functional form of equation [2].<sup>9</sup> Following the above, equation [3] reads as follows.

$$n(t) = \left[a + b\left(\frac{N(t)}{cM(t)}\right)^d\right] \cdot \left[cM(t) - N(t)\right]^e \cdot t \cdot \left[1 + \beta X(t)\right]$$
[3]

X is a vector of macroeconomic variables relevant for the high-yield bond diffusion and  $\beta$  the corresponding vector of parameters. X(t) may reflect current and lagged or carry-over effects on the adoption pattern at time t.

The selection of macroeconomic variables likely to have a bearing on the diffusion process draws on the main determinants of this market underlined by practitioners and academics, which broadly relate corporate financing needs and costs and investor return considerations. According to our review, the high-yield bond market structure is expected to depend mainly on five macroeconomic variables.

The first macroeconomic factor taken into account is leverage buy-outs. In the 1980s high-yield bonds were a controversial instrument in the United States because of their use in financing hostile take-overs and leverage buy-outs (Becketti, 1986, and Taggart, 1988). At that time leverage buy-outs were a major source of primary high-yield bond issuance in the United States, suggesting a positive relationship between leverage buy-outs and high-yield bond diffusion.

The second macroeconomic variable considered is M&A activity. One may expect that high-yield bonds are less frequently used to finance M&A than investment-grade bonds, because notably investment-grade companies acquire other companies. For instance, euro area corporations have tapped the corporate bond market heavily to finance M&A in the first years of the euro (de Bondt, 2002). Hence a negative relationship between high-yield bond diffusion, measured as the percentage of high yield bonds to the overall corporate bond market, and M&A activity is expected.

Given that a major part of revenues obtained from high yield bonds are used to finance investment (Gilson and Warner, 1997) and other business-related financing needs, the third macroeconomic variable taken into account is industrial production. The impact of the business cycle on the diffusion of high-yield bonds could work in both directions. On one hand, when the business cycle is heading upwards, the financing needs of high-yield bond issuers, which are mainly high-growth companies (Yago and Trimbath, 2003), tends to grow above that of other industries. This suggests a positive relationship between high-yield bond diffusion and industrial production. On the other hand, the share of "fallen angels" in the high-yield segment of the corporate bond market rises during economic downturns. This implies a negative relationship between high-yield bond diffusion and industrial production.

The fourth macroeconomic variable taken into account is stock price developments. Movements in the high-yield bond and equity markets are typically positively related. An increase in stock market prices implies an increase in the market value of the firm, which is often perceived both as a form of collateral

<sup>&</sup>lt;sup>9</sup> See Ratkowsky (1990). The mathematical derivation of the logistic curve used by Griliches (1957) and Mansfield (1961) is a bounded continuos time function in which the lower limit is 0 and the upper limit amounts to the market potential. This dampening effect makes sense from an economic perspective but is also analytical needed in order to have an upper limit, since the mathematical solution of the logistic curve (solving a linear differential equation) is a bounded continuos time function.

and as an indication of expected future revenues. Likewise, from an investor perspective, rising stock prices are generally associated to improve economic perspective and lower risk aversion, thereby increasing investors demand for high-yield bonds (Altman, 1992). Consequently, increases in stock market prices would, other things being equal, enhance the demand for junk bonds.

The fifth and final macroeconomic factor considered is the spread between the yield on speculative-grade bonds and BBB-rated bonds, the most risky class of investment-grade bonds. On the one hand, this spread determines the attractiveness for investors to invest in high-yield bonds (demand factor). High-yields are only attractive if they provide, given the perceived credit risk differential, a yield high enough compared with BBB-rated investment-grade bonds. On the other hand, the yield on bonds reflects the (re)financing costs for corporations (supply factor). The higher the costs for corporations to tap the bond market, the less likely they will issue a bond. Which of the factor prevails would depend on the relative attractiveness of alternative financing and investment opportunities.

# 5. Data

The diffusion of financial innovations, and thus also high-yield bonds, can be measured in a number of ways, such as the amounts outstanding in the market, gross/net issues, the number of issues/issuers, and the number of banks or investors participating in the high-yield bond market. This study focuses on the size of the high-yield bond market vis-à-vis the total corporate bond market in terms of the amounts outstanding, both in face and market value terms.<sup>10</sup> The advantage of this measure is, as already pointed out by Berger (1996), that the degree to which financial innovations are adopted is often more important for policy purposes than the number of economic agents participating in the market as Molyneux and Shamroukh (1996) examined.

Chart 5 plots the amounts outstanding of the high-yield bond market in the United States, the United Kingdom and the euro area. The Merrill Lynch data for the United States start in January 1980 and for the United Kingdom and the euro area in December 1997.<sup>11</sup> The chart shows that while the United States market took some time to take off, the European market grew relatively quickly since its inception at the end of 1997. In the United States amounts outstanding started to decline in the late 1980s and recovered after 1993, suggesting a relationship with macroeconomic variables such as the business cycle and stock market developments.

<sup>&</sup>lt;sup>10</sup> Annex 1 provides detailed information on the data. The source used for the bond data is Merrill Lynch, because they have historical information for the United States and Europe and use an equivalent definition in the construction of indices across countries. Their high-yield bond data are based on a composite of Moody's and Standards and Poors which are in national currency and issued by domestic issuers, have at least one year remaining term to maturity, a fixed coupon, and a minimum amount outstanding of EUR 50 million. As regards the coverage of these indices, Merrill Lynch London and New York estimate the coverage for their corporate bond indices in general to be above 90% of the market of publicly issued bonds. This figure has changed over time in the case of the US high-yield index.

<sup>&</sup>lt;sup>11</sup> A drawback of this data, however, is that the US broad high yield index (H0A0) starts in August 1986. Hence we use an earlier but less broad US index (X0A0), which starts in 1980. Although the difference in levels between the two indices is substantial, particularly for the second half of the 1990s, both indices have tended to co-move (see Chart A.1 in Annex 1) and therefore the latter index has been used in the empirical analysis. This index tracks the performance of a selected basket of the most liquid below-investment grade US dollar-denominated corporate bonds publicly issued in the United States domestic market.

# Chart 5 Amounts outstanding of high-yield bonds in the United States, United Kingdom and the euro area

(United States in US dollar millions (left-hand scale); United Kingdom in British pound millions and euro area in euro millions (right-hand scale); the continuous lines represent face value, the dotted lines represent market value)



Chart 6 shows the development of the high-yield bond market measured in amounts outstanding as a proportion of the overall corporate bond market. Starting from a very low base, the market share of the high-yield bond market in the United States grew very fast during the 1980s, but subdued in the early 1990s. It stabilised in the second part of the 1990s. In the United Kingdom, the high-yield segment of the corporate bond market grew quicker than in the euro area, probably benefiting from having a more market-based financial structure. The share of the euro area high-yield bond market as a share of the overall corporate bond market grew steadily since the beginning of 1998, and stabilised in 2001. From a modelling perspective this would suggest a faster speed of diffusion, as measured by the coefficient b, in Europe than in the United States.



Two potential determinants of high-yield bonds issuance are leverage buy-outs and M&A. Chart 7 indicates that United States leverage buy-outs reached extraordinary peaks in the 1985-1989 period, and declined sharply during the early 1990s. The United Kingdom, the United States and the euro area show an increase in leverage buy-outs in the second part of the 1990s. In the United States the value of leverage buy-outs declined again in 2001, whereas the European figures, although at a lower level, remained similar to 2000 values. As shown in Chart 8, the United States has experienced a fast process of industrial concentration, as reflected in M&A, which accelerated substantially in the second half of the 1990s. This process declined with the slowdown on economic activity in 2001. On a much lower scale, during the 1980s there was an increase in the amount of M&As which peaked in the late 1980s in the United States (Altman and Smith, 1991). A similar pattern of a peak in M&A around 2000, albeit on a lower base, is visible for the United Kingdom and the euro area.



20 ECB Working Paper Series No. 313 February 2004

#### 6. **Empirical results**

The empirical results are based on a diffusion period for the United States and for the United Kingdom and the euro area on a diffusion period and a full sample period, which ends in December 2001. The diffusion periods are defined as the period from the start of the high-yield bond market up to the peak of the market penetration level, e.g. the maximum share of high-yield bonds in the total corporate bond market.<sup>12</sup> For the United States the diffusion period ends in August 1988, well before Drexel Burnham Lambert declared bankruptcy in early 1990. For the United Kingdom and the euro area the diffusion period ends in July 2000 and January 2001, respectively. Since we can not yet be sure whether the peaks in the high-yield market penetration as observed for the United Kingdom and the euro area are a local or absolute maximum also full-sample results are shown for Europe. In other words, the short history of these high-vield bond markets stresses the need to interpret the European empirical results with more than usual caution.

#### 6.1 **Empirical diffusion models**

Our empirical analysis starts with a simple diffusion model, because increasing the model complexity might increase the bias in the estimates of the diffusion parameters (Van de Bulte and Lilien, 1997). Consequently, our starting point is the one-parameter diffusion model introduced by Mansfield (1961), assuming that in equation [2] and [3] parameters c and e equals 0.15 and 1, respectively. The value of 0.15 for the market ceiling or market penetration level is based on the maximum value achieved of 0.10for the US high-yield market share, MS, plus two standard deviations. The results for the diffusion parameters presented in this paper appear, however, to be rather insensitive of using different values for the market potential, at least as set above any observed market penetration level (see Annex 3). It is tested whether the one-parameter diffusion model (Mansfield model) can be extended to a two-parameter diffusion model (Bass model,  $a \neq 0$ ) or even a three-parameter model (non-uniform influence (NUI) model,  $a \neq 0$  and  $d \neq 1$ ). Our focus is on the annual rate of diffusion and we refer to Annex 1 for the notation of the variables. This results in the following empirical diffusion model, including macroeconomic variables, which has been estimated by non-linear least squares.<sup>13</sup>

$$\Delta_{12}MS_{t} = \left[a + b\left(\frac{MS_{t}}{0.15}\right)^{d}\right] \cdot \left[0.15 - MS_{t}\right]^{1} \cdot t \cdot \left[1 + \beta_{1}\ln(\sum_{i=0}^{2} LBNV_{t-3-i}) + \beta_{2}\ln(\sum_{i=0}^{11} MANV_{t-3-i} + \beta_{3}\frac{1}{3}\sum_{i=0}^{2} IPIN_{t-i} + \beta_{4}\frac{1}{3}\sum_{i=0}^{2} SMIN_{t-i} + \beta_{5}\Delta_{12}(hyyi - bbbyi)_{t}\right]$$

$$(4)$$
With:

with

<sup>&</sup>lt;sup>12</sup> Van den Bulte and Lilien (1997) show that there is no general understanding of the number of data required to obtain stable parameter estimates and whether this number typically occurs prior to or after the peak in adoption. For the reason that estimates of diffusion model coefficients may change as one extends the number of observations, the sample is broken at the peak in the high-yield bond adoption. The maximum Chow breakpoint F-statistic, based on the standard Mansfield and Bass models, is reached in April 1988 in the United States, in October 2000 in the United Kingdom, and in March 2001 in the euro area. The breakpoint chosen on the basis of the maximum high-yield bond market share is four months later for the United States, three months earlier for the United Kingdom, and two months earlier for the euro area.

$$MS_{t} = \frac{HYFV_{t}}{TCFY_{t}} \quad or \quad \frac{HYMV_{t}}{TCMV_{t}}$$
$$\dot{X}_{t} = 100 \cdot \frac{X_{t} - X_{t-12}}{X_{t-12}}$$

The lag structure of M&A and the industrial production growth is in line with the empirical findings of de Bondt (2002) for the euro area. This study shows that M&A financing needs affects bond financing up to one year and industrial production up to one quarter. Given the fact that bridge financing is less likely for leverage buy-outs compared with M&A, a quarterly instead of annual flow of leverage buy-outs is considered. In addition, it is assumed that it typically takes one quarter before the financing of leverage buy-outs and M&A is settled. The lag structure of the annual stock market return is identical to that of the annual growth of industrial production, that is a quarterly moving average. The high-yield spread is expected to have an immediate impact, given that it reflects the forward looking expectations of expected credit risk, taking into acount the price of the quantity developments we want to explain.

### 6.2 Estimates

The estimates of equation [4] appear to be rather insensitive to whether the high-yield bond diffusion is measured in face or market value terms (compare Table 4 with Table 5). Six main findings emerge from Table 4 and 5.

First, the high-yield bond diffusion rate is significant in all cases except one. The next section examines in detail the estimated diffusion rates. For the United States also a pioneer influence factor or rate of innovation is typically found to be significant. The United States was the innovator of junk bonds and the high-yield bond technology became therefore gradually available, while for Europe all information concerning the high-yield bond innovation was, as a late adopter, immediately available at the beginning of the diffusion period. Consequently, the pioneer influence factor is found to be important in the United States, but not in Europe.

The second empirical finding is that leverage buy-outs have had a significantly positive impact on the high-yield bond diffusion. An increase in leverage buy-outs by one million has resulted in a rise in the high-yield bond market share of around 0.01 percentage points in the United States and the euro area. This suggests that high-yield bonds have been issued to finance leverage buy-outs, confirming the US debate in the 1980s of buoyant junk bond financing of leverage buy-outs.

Third, M&A activity has significantly negatively affected the high-yield bond diffusion process. An increase in the annual flow of M&A by one million has resulted in a decrease in the high-yield bond share in all corporate bonds of around 0.08 percentage points in the United States and Europe. This result

<sup>&</sup>lt;sup>13</sup> This estimation method is in line with Srinivasa and Mason (1986) and Hansen et al. (2003), which broadly show that a relatively simple method such as non-linear least squares tend to be very close to those obtained with more complicated estimation procedures.

indicates that high-yield bonds were less frequently used to finance M&A than investment-grade bonds, in line with the fact that typically investment-grade companies acquire speculative-grade companies.

The fourth finding is that the output impact of fallen angels seems to have been dominant in the United States during the high-yield bond diffusion period, but not in Europe. A one-percentage point increase in industrial production growth has resulted in the United States to a decrease of around 0.3 percentage points in the share of high-yield bonds in the total corporate bond market, whereas it has led in Europe to an increase by up to 2 percentage points. This finding suggests that the high-yield bond market in Europe has been particularly important for growth companies.

Fifth, stock market developments have significantly positively affected the high-yield bond diffusion. The impact of the stock market return on the high-yield bond diffusion process is found to be larger in the United States and the United Kingdom than in the euro area. Broadly speaking, a one-percentage point increase in the stock market return leads to a rise in the share of high-yield bonds in all corporate bonds of between 0.1 and 0.2 percentage points in the United States and the United Kingdom and 0.05 percentage points in the euro area. This finding may be related to the fact that the stock market capitalisation in the United States and the United Kingdom is higher than in the euro area.

The sixth and last finding is that the spread between the yield on speculative-grade bonds and BBB-rated investment-grade is found to have plaid an insignificant role in the high-yield bond diffusion process in the United States and the euro area. In the United Kingdom, however, this high-yield spread has significantly positively affected the high-yield bond diffusion process. This suggests that a strong demand from high-yield bond investors, searching for higher yields, have been dominant in the United Kingdom during this period.

In sum, the macroeconomic conditions do play a significant role for the high-yield bond diffusion (see also the Wald tests in Table 4 and 5). The models including macroeconomic variables are therefore preferred above the models without any macroeconomic effects.

Pioneer influence factor	Diffusion speed	LBO	M&A	Output growth	Stock market return	High-yield spread	Wald test Bass/NUI model <sup>1)</sup>	Wald test macro factors <sup>2)</sup>	R <sup>2</sup>
a	b	β <sub>1</sub>	β <sub>2</sub>	β <sub>3</sub>	β <sub>4</sub>	β <sub>5</sub>			
United Sta	tes 1981.1-1	988.8 <sup>3)</sup>							
	0.507**						6.94**		0.73
	(0.04)								
-0.021**	0.578**						5.80** <sup>4)</sup>		0.76
(0.01)	(0.07)								
	0.053*	0.007	-0.082**	-0.318	0.243	-0.734	11.3**	607**	0.85
	(0.02)	(0.01)	(0.01)	(0.32)	(0.14)	(0.93)			
-0.009**	0.170**	0.006**	-0.086**	-0.260*	0.085**	-0.146	5.51* <sup>4)</sup>	6887**	0.87
(0.00)	(0.04)	(0.00)	(0.00)	(0.11)	(0.03)	(0.26)			
United Kir	ngdom 1988.	12-2000.7							
	0.465**						6.06* <sup>5)</sup>		0.00
	(0.07)								
	0.495**	0.001	-0.084**	2.258**	0.177*	3.066**	0.57	11224**	0.53
	(0.06)	(0.01)	(0.01)	(0.30)	(0.07)	(0.49)			
United Kir	1988. ngdom	12-2001.12							
	0.233**						7.60** <sup>5)</sup>		0.00
	(0.09)								
	0.337**	-0.004	-0.077**	1.174**	0.047	0.241	1.70	12290**	0.42
	(0.17)	(0.01)	(0.01)	(0.26)	(0.07)	(0.65)			
Euro area	1998.12-200	1.1							
	0.603**						0.43		0.75
	(0.05)								
	0.335**	0.012**	-0.083**	0.574**	0.050	-0.119	0.68	38566**	0.95
	(0.07)	(0.00)	(0.00)	(0.12)	(0.03)	(0.15)			
Euro area	1998.12-200	1.12							
	0.389**						3.32		0.00
	(0.10)								
	0.329**	0.007*	-0.080**	0.755**	0.042	-0.005	5.30*	97695**	0.92
	(0.08)	(0.00)	(0.00)	(0.11)	(0.04)	(0.15)			
0.011*	0.192*	0.010**	-0.082**	0.942**	0.048	-0.097	0.91	53147**	0.93
(0.00)	(0.08)	(0.00)	(0.00)	(0.18)	(0.04)	(0.20)			

 Table 4 Estimation results of high-yield bond diffusion in face value terms

Notes: heteroscedasticity and autocorrelation-corrected standard errors between parentheses; \*\* and \* denote significance at the 1% and 5% level, respectively; <sup>1</sup>) F-statistic, insignificance denotes that a more extended diffusion model is not rejected against the estimated diffusion model, that is, the Bass against Mansfield model (test a=0 in Bass model) and the NUI against Bass model (test a=0 and d=1 in NUI model); <sup>2</sup>) F-statistic; significance denotes that macroeconomic effects are jointly significantly different from zero; <sup>3</sup>) starting in 1981.03 for diffusion models including macroeconomic variables; <sup>4</sup>) non-uniform influence factor, d, however, not significantly different from 0; <sup>5</sup>) b < 0, however.

Sources: various and authors' estimates.



Pioneer influence factor	Diffusion speed	LBO	M&A	Output growth	Stock market return	High-yield spread	Wald test Bass/NUI model <sup>1)</sup>	Wald test macro factors <sup>2)</sup>	R <sup>2</sup>
a	b	β1	β <sub>2</sub>	β <sub>3</sub>	β4	β <sub>5</sub>			
United Sta	tes 1981.1-1	988.8 <sup>3)</sup>							
	0.490**						4.15*		0.66
	(0.05)								
-0.019*	0.554**						3.25		0.69
(0.01)	(0.08)								
	0.046*	0.006	-0.083**	0.245	0.357*	-1.075	6.56**	501**	0.82
	(0.02)	(0.01)	(0.01)	(0.44)	(0.19)	(1.35)			
-0.007*	0.151**	0.006	-0.087**	-0.046	0.126**	-0.320	2.15	5990**	0.84
(0.00)	(0.04)	(0.00)	(0.00)	(0.12)	(0.05)	(0.44)			
United Kir	ngdom 1998.	.12-2000.7							
	0.454**						5.04* <sup>4)</sup>		0.00
	(0.05)								
	0.440**	0.001	-0.083**	2.251**	0.175*	2.643**	1.01	6214**	0.30
	(0.06)	(0.01)	(0.01)	(0.33)	(0.08)	(0.57)			
United Kir	ngdom 1998.	.12-2001.12							
	0.210*						1.10		0.04
	(0.09)								
	0.273	-0.006	-0.074**	1.252**	0.040	-0.438	0.55	11369**	0.59
	(0.15)	(0.01)	(0.01)	(0.30)	(0.07)	(1.02)			
Euro area	1998.12-200	01.1							
	0.596**						0.03		0.53
	(0.08)								
	0.318**	0.010*	-0.080**	0.426**	0.055*	-0.610**	0.35	103003**	0.97
	(0.06)	(0.00)	(0.00)	(0.12)	(0.03)	(0.20)			
Euro area	1998.12-200	1.12							
	0.339*						0.01		0.05
	(0.14)								
	0.373**	0.005*	-0.079**	0.645**	0.047	-0.376	0.18	206901**	0.96
	(0.09)	(0.00)	(0.00)	(0.08)	(0.03)	(0.23)			

Table 5 Estimation results of high-yield bond diffusion in market value terms

Notes: heteroscedasticity and autocorrelation-corrected standard errors between parentheses; \*\* and \* denote significance at the 1% and 5% level, respectively; <sup>1)</sup> F-statistic, insignificance denotes that a more extended diffusion model is not rejected against the estimated diffusion model, that is, the Bass against Mansfield model (test a=0 in Bass model) and the NUI against Bass model (test a=0 and d=1 in NUI model); <sup>2)</sup> F-statistic; significance denotes that macroeconomic effects are jointly significantly different from zero; <sup>3)</sup> starting in 1981.03 for diffusion models including macroeconomic variables; <sup>4)</sup> b < 0, however.

Sources: various and authors' estimates.

### 6.3 Comparison of high-yield bond diffusion rates

Two main conclusions emerge when comparing the US results for the estimated autonomous speed of diffusion in an illustrative way with those of the United Kingdom and the euro area (see Table 6).

First, the diffusion rate in the United States seems to have been more sensitive to macroeconomic developments than in Europe. This may be related to the pioneer stage of high-yield bond diffusion in the United States during the period under review. Uncertainty about the longer-term viability of junk bonds may have resulted in a relatively strong response to the macroeconomic environment. Second, the diffusion rate is found to be slower in the United States than in Europe, at least as macroeconomic factors are taken into account. This suggests a much slower high-yield bond diffusion pattern in the United States than in the United States during the period under review.

Diffusion models	United	United	United	Euro area	Euro area
	States	Kingdom	Kingdom		
	1981.1-	1998.12-	1998.12-	1998.12-	1998.12-
	1988.8	2000.7	2001.12	2001.1	2001.12
Face value					
Mansfield	0.51	0.47	0.23	0.60	0.39
Bass	0.58	_	_	_	_
Mansfield with macroeconomic variables	0.05	0.50	0.34	0.34	0.33
Bass model with macroeconomic variables	0.17	_	_	_	0.19
Market value					
Mansfield	0.49	0.45	0.21	0.60	0.34
Bass	0.55	_	_	_	_
Mansfield with macroeconomic variables	0.05	0.44	0.27	0.32	0.37
Bass model with macroeconomic variables	0.15	_	—	_	_

### Table 6 High-yield bond diffusion rates in the United States, United Kingdom and the euro area

*Note: figures are based on the estimation results of Table 4 and 5.* 

Sources: various and authors' estimates.

Given the fact that when modelling a financial innovation the corresponding market is in a transition stage, a stage of innovation and imitation, the diffusion rate using the classic Mansfield model is estimated repeatedly, using ever larger subsets of the sample data. The first observation is used to form the first estimate of the diffusion rate. The next observation is then added to the data set to compute the second estimate of the diffusion rate. This process is repeated until all observations up to December 2001 have been used (see Chart 9). Two observations emerge from this.

The first observation is that the coefficient of imitation or the diffusion rate increases as observations are added up to a certain point for the United States and the euro area, after which the imitation rate tends towards zero. For the United Kingdom the estimated diffusion rate declines steadily as we add more data over the sample period, approaching a value of zero. The tendency to decline towards zero is exactly what one expects given the fact that the diffusion of a financial innovation is expected to take place only for a limited time span. For example, the classic diffusion periods as used in the diffusion literature for black & white televisions and colour televisions are 1947-1953 and 1963-1970, respectively. These diffusion periods matches well with the length of the diffusion period for high-yield bonds in the United States of 1981-1988, as defined in this study. After the diffusion of a financial innovation is completed no longer

an impact of the autonomous speed of diffusion is expected. Given the tendency to decline for the UK and euro area diffusion rates, Chart 9 suggests that the diffusion period has completed in Europe.

Secondly, it seems to be the case that the fastest diffusion speed of around 0.6 reached for the United States forms more or less the starting point for the high-yield bond diffusion in Europe. The comparatively fast speed of diffusion in Europe can be attributed to two factors. First and foremost, it could be due to the fact that there is a "bandwagon effect", that is the junk bond market was a US innovation that has since spilled over to the European market. Moreover, most of the main underwriters in Europe are US banks (see Table 3 in Section 2), which had already acquired the pricing, underwriting and placement expertise in their home country (Harm, 2001). More generally, a mechanism of financial innovation diffusion across countries may exist. Gatignon et al. (1989) and Dekimpe et al. (1996) show that a so-called "left-hand truncation bias" exists: later adopting countries show a relatively quick diffusion rate. The country that "imports" the financial innovation, i.e. Europe, adopts much faster than the pioneer country, i.e. the United States, because late adopters are in a better position to assess the new technology than earlier ones since all or at least more information concerning the innovation is available. Finally, improvement in computers, telecommunications, and other advances in technology coupled with financial de-regulation may have all contributed to speeding up the diffusion of financial innovations and shortening the innovation and imitation cycle (Merton, 1995).

### **Chart 9 Recursive estimates of Mansfield diffusion rates**

(grey lines denote two standard errors confidence interval)



### **United States**

### Euro area



Sources: Merrill Lynch and authors' estimates.

## 7. Concluding remarks

The existence of a mature corporate bond market including a deep high-yield bond segment appears to be positive for economic development. It allows many corporations to raise funds more quickly and on more flexible terms than would otherwise have been available solely from the banking sector. We believe that the existence of this avenue of corporate finance complementary to the finance provided by the loan and equity market is particularly beneficial for Europe due to the existence of a large number of small and medium sized firms.

Unlike in the United States, where the high yield bond market developed in the early 1980s, in Europe the high-yield segment of the corporate bond market is a phenomenon of the late 1990s. The observed quick diffusion of high-yield bonds in Europe suggests that a national financial landscape might change within a short time span, because of a quick adoption of financial innovations from a pioneer country, e.g. the United States, to later adopting countries, e.g. the United Kingdom and euro area countries.

From a modelling point of view, this study provides a framework, taking account of the macroeconomic environment, which may form the starting point for future research on the diffusion of other financial innovative products, for instance all type of securitised assets. Another promising avenue for future research is the diffusion of financial instruments from industrialised economies to emerging market economies, since industrial economies appear, in contrast to emerging market economies, to benefit from the existence of multiple avenues of corporate finance (Davis and Stone, 2003). Corporations from emerging markets, which often have a credit quality below investment grade, could, however, also benefit from tapping the bond market when corporate profits are under pressure or when banks cut back their lending. High-yield bond markets could thus play a very important role in the development of financial markets in emerging market zones and therefore economic growth.

# **ANNEX 1 Data description**

Code	Description	Unit <sup>1)</sup>	Source
USHYFV	US High Yield 175 (X0A0), face value	US dollar millions, end of month	Merrill Lynch
UKHYFV	UK High Yield (HL00), face value	Pound millions, end of month	Merrill Lynch
EAHYFV	Euro High Yield (HE00), face value	Euro millions, end of month	Merrill Lynch
USHYFV2	US High Yield Master II (H0A0), face value	US dollar millions, end of month	Merrill Lynch
UKHYMV	UK High Yield (HL00), market value	Pound millions, end of month	Merrill Lynch
EAHYMV	Euro High Yield (HE00), market value	Euro millions, end of month	Merrill Lynch
USHYMV	US High Yield 175 (X0A0), market value	US dollar millions, end of month	Merrill Lynch
USHYMV2	US High Yield Master II (H0A0), market value	US dollar millions, end of month	Merrill Lynch
USTCFV	US total corporate (C0A0), face value	US dollar millions, end of month	Merrill Lynch
UKTCFV	UK total corporate (UC00), face value	Pound millions, end of month	Merrill Lynch
EATCFV	Euro total corporate (ER00), face value	Euro millions, end of month	Merrill Lynch
USTCMV	US total corporate (C0A0), market value	US dollar millions, end of month	Merrill Lynch
UKTCMV	UK total corporate (UC00), market value	Pound millions, end of month	Merrill Lynch
EATCMV	Euro total corporate (ER00), market value	Euro millions, end of month	Merrill Lynch
USHYYI	Yield on US high-yield bonds (XOA0)	% per annum, end of month	Merrill Lynch
UKHYYI	Yield on UK high-yield bonds	% per annum, end of month	Merrill Lynch
EAHYYI	Yield on euro area high-yield bonds (HE00)	% per annum, end of month	Merrill Lynch
USBBBYI	Yield on US BBB-rated investment-grade bonds (C0A4)	% per annum, end of month	Merrill Lynch
UKBBBYI	Yield on UK BBB-rated investment-grade bonds (UC40)	% per annum, end of month	Merrill Lynch
EABBBYI	Yield on euro area BBB-rated investment-grade bonds (ER40)	% per annum, end of month	Merrill Lynch
USMANV	Nominal value of effective M&A deals involving US non-	US dollar millions, monthly flows	Thomson Financial
	financial corporations as targets	· · · ·	
UKMANV	Nominal value of effective M&A deals involving UK non-	British pound millions, monthly	Thomson Financial
<b>E</b> + <b>X</b> +	financial corporations as targets	flows	TI F. I
EAMANV	Nominal value of effective M&A deals involving euro area non-	EUR millions, monthly flows	Thomson Financial
USIDNV	Nominal value of effective leverage buy outs deals involving US	US dollar millions monthly flows	Thomson Financial
USLBINV	non-financial corporations as targets	0.5 donar minions, montiny nows	Thomson Financial
UKLBNV	Nominal value of effective leverage buy-outs deals involving UK	British pound millions, monthly	Thomson Financial
CHEDICO	non-financial corporations as targets	flows	
EALBNV	Nominal value of effective leverage buy-outs deals involving	EUR millions, monthly flows	Thomson Financial
	euro area non-financial corporations as targets		
USIPIN	US industrial production index	1992=100	Federal Reserve
UKIPIN	UK industrial production index	1995=100	National Statistical Office
EAIPIN	Euro area industrial production index	1995=100	Eurostat
EAIPAG	Euro area industrial production, adjusted for working days	Annual percentage changes in three-	Eurostat
		month moving averages	<b>D</b> i i
USSMIN	US stock market index, Standard & Poor's 500	19/3 = 100, end of month	Datastream
UKSMIN	UK stock market index, Datastream UK index	19/3 = 100, end of month	Datastream
EASMIN	Euro area stock market index, Dow Jones EURO STOXX broad	19/3=100, end-of-month	Datastream

1) Euro is euro or euro legacy currencies.

## Chart 1.1 Comparison of the two US high-yield bond indices calculated by Merrill Lynch

(US dollar millions; broad index is on the left-hand scale, historical index is on the right-hand scale)



Source: Merrill Lynch.

Working Paper Series No. 313 February 2004

# **ANNEX 2** Diffusion patterns according to different values for the Bass diffusion model parameters

The impact of different values for the estimated diffusion model parameters is graphically shown below for Bass models with a market penetration ceiling of 15%. Coefficient a captures the role of external adopters, innovators or pioneers. This pioneer influence factor does not affect the overall shape of the adoption curve, but does affect the intercept, and therefore the number of periods necessary needed to reach the peak (see Chart 2.1). Parameter b reflects the role of internal adopters or imitators. The larger this autonomous speed of diffusion or rate of imitation, the more rapid the new product diffusion. The diffusion parameter b affects both the timing and amplitude of the peak (see Chart 2.2).



# **ANNEX 3 Estimation results of high-yield bond diffusion based on different** market penetration levels

An exogenous set market potential or market penetration ceiling is a common practice and the value of 15%, as used in the main text, is based on the actual observed maximum plus two standard deviations. Nevertheless, Table 2a and 2b below show the sensitivity of our results to changes in the market potential to 6% (the level around which the US market share of high-yield bonds fluctuates since the early 1990s) and to 10% above 15%, that is 25%. Table A.3a provides economically and statistically implausible results for the United States, simply due to the fact that the maximum market penetration level is assumed to be below observed values, which is inappropriate in a diffusion modelling framework. Comparing the other results as reported in Table A.3a and Table A.3b with Table 4 in the main text, one may conclude that the results remains very similar for other values of the market potential as long as the assumed potential market share is above any observed market share.

Pioneer influence factor	Diffusion rate	LBO	M&A	Output growth	Stock market return	High-yield spread	Wald test Bass/NUI model <sup>1)</sup>	Wald test macro factors <sup>2)</sup>	R <sup>2</sup>
a	b	β1	β <sub>2</sub>	β <sub>3</sub>	β4	β <sub>5</sub>			
United Sta	tes 1981.1-1	<b>988.8</b> <sup>3)</sup>							_
	-0.196						17.8**		0.00
	(0.11)								
0.166**	-0.406**						0.33		0.00
(0.04)	(0.12)								
	0.103*	-0.065	-0.040	0.522	-0.180*	2.327	18.0**	5448**	0.00
	(0.04)	(0.03)	(0.02)	(0.69)	(0.09)	(2.29)			
-0.077**	0.474**	-0.020**	-0.071**	0.062	-0.050**	1.108*	0.84	150705**	0.07
(0.02)	(0.11)	(0.01)	(0.00)	(0.22)	(0.02)	(0.50)			
United Kir	ngdom 1998.	.12-2000.7							
	1.097**						0.28		0.04
	(0.14)								
	0.779**	0.021	-0.097**	2.829**	0.126	3.014**	0.23	5590**	0.53
	(0.15)	(0.02)	(0.01)	(0.53)	(0.07)	(0.61)			
United Kir	ngdom 1998.	.12-2001.12							
	0.619**						5.83** <sup>4)</sup>		0.05
	(0.21)								
	0.758*	0.004	-0.083**	1.420**	0.028	0.315	1.15	7003**	0.46
	(0.35)	(0.02)	(0.01)	(0.40)	(0.09)	(0.69)			
Euro area	1998.12-200	)1.1							
	0.879**						11.0**		0.66
	(0.07)								
-0.111**	1.198**						0.11		0.81
(0.03)	(0.11)								
	0.414**	0.015**	-0.085**	1.018**	0.038	-0.078	0.00	12391**	0.95
	(0.08)	(0.00)	(0.00)	(0.16)	(0.03)	(0.15)			
Euro area	1998.12 200	1.12							
	0.611**						0.07		0.08
	(0.14)								
	0.409**	0.010**	-0.082**	0.986**	0.039	-0.015	0.01	75024**	0.94
	(0.09)	(0.00)	(0.00)	(0.10)	(0.04)	(0.13)			

Table A.3a Estimation results of high-yield bond diffusion in face value terms, based on a maximum market penetration level of 6%

Notes: heteroscedasticity and autocorrelation-corrected standard errors between parentheses; \*\* and \* denote significance at the 1% and 5% level, respectively; <sup>1)</sup> F-statistic, insignificance denotes that a more extended diffusion model is not rejected against the estimated diffusion model, that is, the Bass against Mansfield model (test a=0 in Bass model) and the NUI against Bass model (test a=0 and d=1 in NUI model); <sup>2)</sup> F-statistic; significance denotes that macroeconomic effects are jointly significantly different from zero; <sup>3)</sup> starting in 1981.03 for diffusion models including macroeconomic variables; <sup>4)</sup> b < 0, however.

Sources: various and authors' estimations.

Pioneer influence factor	Diffusion rate	LBO	M&A	Output growth	Stock market return	High-yield spread	Wald test Bass/NUI model <sup>1)</sup>	Wald test macro factors <sup>2)</sup>	R <sup>2</sup>
a	b	β1	β <sub>2</sub>	β <sub>3</sub>	β4	β5			
United Sta	ntes 1091.1-1	998.8 <sup>3)</sup>							
	0.368**						0.01		0.65
	(0.04)								
	0.071**	0.004	-0.083**	-0.433*	0.130*	-0.335	6.66**	3996**	0.82
	(0.02)	(0.00)	(0.00)	(0.20)	(0.06)	(0.56)			
-0.004*	0.148**	0.004	-0.084**	-0.334**	0.068**	-0.049	4.35* <sup>4)</sup>	12852**	0.84
(0.02)	(0.04)	(0.00)	(0.00)	(0.11)	(0.03)	(0.27)			
United Kin	ngdom 1998.	12-2000.7							
	0.400**						6.85** <sup>5)</sup>		0.00
	(0.06)								
	0.443**	-0.000	-0.082**	2.227**	0.181*	3.105**	0.90	10955**	0.52
	(0.06)	(0.01)	(0.01)	(0.30)	(0.08)	(0.53)			
United Ki	ngdom 1998.	12-2001.12							
	0.199**						7.38** <sup>5)</sup>		0.00
	(0.07)								
	0.292*	-0.004	-0.076**	1.154**	0.049	0.232	1.72	12924**	0.41
	(0.15)	(0.01)	(0.01)	(0.25)	(0.07)	(0.65)			
Euro area	1998.12-200	1.1							
	0.554**						0.04		0.74
	(0.05)								
	0.320**	0.012**	-0.083**	0.510**	0.047*	-0.120	1.09	47642**	0.94
	(0.07)	(0.00)	(0.00)	(0.11)	(0.03)	(0.16)			
Euro area	1998.12-200	1.12							
	0.354**						4.37*		0.00
	(0.09)								
	0.310**	0.006**	-0.080**	0.727**	0.042	-0.002	6.94*	90572**	0.92
	(0.08)	(0.00)	(0.00)	(0.11)	(0.04)	(0.16)			
0.007*	0.157*	0.010*	-0.082**	0.949**	0.049	-0.113	0.69	49497**	0.93
(0.03)	(0.07)	(0.00)	(0.00)	(0.18)	(0.05)	(0.21)			

Table A.3b Estimation results of high-yield bond diffusion in face value terms, based on a maximum market penetration level of 25%

Notes: heteroscedasticity and autocorrelation-corrected standard errors between parentheses; \*\* and \* denote significance at the 1% and 5% level, respectively; <sup>1)</sup> F-statistic, insignificance denotes that a more extended diffusion model is not rejected against the estimated diffusion model, that is, the Bass against Mansfield model (test a=0 in Bass model) and the NUI against Bass model (test a=0 and d=1 in NUI model); <sup>2)</sup> F-statistic; significance denotes that macroeconomic effects are jointly significantly different from zero; <sup>3)</sup> starting in 1981.03 for diffusion models including macroeconomic variables; <sup>4)</sup> non-uniform influence factor, d, however, not significantly different from 0; <sup>5)</sup> b < 0, however.

Sources: various and authors' estimations.



## References

Allen, F. and D. Gale, 1994, Financial Innovation and Risk Sharing, MIT Press, Cambridge MA.

Allen F. and D. Gale, 2000, Comparing Financial Systems, MIT Press, Cambridge MA.

Altman, E.I., 1992, Revisiting the high-yield bond market, Financial Management, 21, 2, 78-92.

Altman, E.I., 1998, The anatomy of the high yield bond market: after two decades of activity-implications for Europe, *Salomon Smith Barney, Global Corporate Bond Research Group*, September.

Altman, E.I. and R.C. Smith, 1991, Highly leveraged restructurings: a valid role for Europe, *Journal of International Securities Markets*, 347-358.

Bass, F.M., 1969, A new product growth model for consumer durables, *Management Science*, 15, 215-227.

Bass, F.M., 1980, The relationship between diffusion rates, experience curves and demand elasticities for consumer durable technological innvations, *Journal of Business*, 53, 2, 51-67.

Bass, F.M., T.V. Krishan, D.C. Jain, 1994, Why the Bass model fits without decision variables, *Marketing Science*, 13, 3, 203-223.

Becketti, S., 1986, Corporate mergers and the business cycle, *Federal Reserve Bank of Kansas City Economic Review*, 71, May, 13-26.

Berger, A.N., 1996, Comment on diffusion of financial innovations: the case of junk bonds and note issuance facilities, *Journal of Money, Credit, and Banking*, 28, 3, 523-526.

BIS, 2001, The changing shape of fixed income markets, *BIS Working Papers*, Study Group on Fixed Income Markets, 104.

Bondt, G.J. de, 2000, Off-balance-sheet activities in Europe: empirical evidence from bank-level panel data, in: G.J. de Bondt, 2000, *Financial Structure and Monetary Transmission in Europe: A Cross-Country Study*, Edward Elgar, Cheltenham, UK / Northampton, MA, USA, 46-64.

Bondt, G. de, 2002, Euro area corporate debt securities market: first empirical evidence, *ECB Working Paper*, 164.

Brewer, E. and W.E. Jackson, 2000, Requiem for a market maker: the case of Drexel Burnham Lambert and junk bonds, *Journal of Financial Services Research*, 17, 3, 209-235.

Bulte, C. van den, and G.L. Lilien, 1997, Bias and systematic change in the parameter estimates of macro-level diffusion models, *Marketing Science*, 16, 4, 338-353.

Cooper, N., R. Hillman and D. Lynch, 2001, Interpreting movements in high-yield corporate bond market spreads, *Bank of England Quarterly Bulletin*, 41, 110-120.

Davies, S., 1979, Diffusion of Process Innovations, Cambridge University Press.

Davis, E.P., 2001, Multiple avenues of intermediation, corporate finance and financial stability, *IMF Working Paper*, No.01/115.

Davis, E.P. and M. Stone, 2003, Corporate financial structure and financial stability, *Paper presented at a Birmingham University seminar*, June.

Dekimpe, M.G., P.M. Parker and M. Sarvary, 1996, Comparing adoption patterns: a global approach, *INSEAD R&D Working Paper*, 96/37/MKT.

DeRosa-Farag, S. and J. Blau, 1999, Changes in the high yield market: a historic perspective, in: T.M. Barnhill Jr., W.F. Maxwell and M.R. Shenkman (ed.), *High-Yield Bonds: Market Structure, Portfolio Management and Credit Risk Modelling* 17-39,McGraw-Hill.

Easingwood, C.J., V. Mahajan and E. Muller, 1983, A nonuniform influence innovation diffusion model of new product acceptance, *Marketing Science*, 2, 3, 273-295.

ECB, 2000, The euro area one year after the introduction of the euro: key characteristics and changes in the financial structure, *Monthly Bulletin*, January, 35-49.

ECB, 2001, Characteristics of corporate finance in the euro area, Monthly Bulletin, February, 37-50.

Fourt, L.A. and J.W. Woodlock, 1960, Early prediction of market success for grocery products, *Journal of Marketing*, 25, 31-38.

Gatignon, H., J. Eliashberg and T.S. Robertson, 1989, Modeling multinational diffusion patterns: an efficient methodology, *Marketing Science*, 8, 3, 231-247.

Garman, M.C. and M.S. Fridson, 1996, Monetary influences on the high yield spread versus Treasuries, *Extra Credit*, July/August, 13-26.

Geroski, P.A., 1999, Models of technology diffusion, *Centre for Economic Policy Research Discussion Paper*, 2146.

Gertler, M. and C. Lown, 1999, The information in the high-yield bond spread for the business cycle: evidence and some implications, *Oxford Review of Economic Policy*, 15, 3, 132-150.

Gilson, S.C. and J.B. Warner, 1997, Junk bonds, bank debt, and financing corporate growth, *Harvard Business School and University of Rochester Mimeo*.

Goldsmith, R., 1969, Financial Structure and Development, Yale University Pres.

Griliches, Z., 1957, Hybrid corn: an exploration in the economics of technical change, *Econometrica* 25, 501-522.

Harm, C., 2001, European financial market integration: the case of private sector bond and syndicated loans, *Journal of International Financial Markets, Institutions and Money*, 11, 245-263.

Gruber, H. and F. Verboven, 2001, The diffusion of mobile telecommunications services in the European Union, *European Economic Review*, 45, 3, 577-588.

Hansen, B., M. Cortina-Borja and S.G. Ratcliffe, 2003, Assessing non-linear estimation procedures for human growth models, *Annals of Human Biology*, 30, 1, 80-96.

Harm, C., 2001, European financial market integration: the case of private sector bonds and syndicated loans, *Journal of International Financial Markets, Institutions and Money*, 11, 3-4, 245-263.

Herring, R.J. and N. Chatustipitak, 2001, The case of the missing market: the bond market and why it matters for financial development, *Wharton School Working Paper*, 01-08.

Jagtiani, J., A. Saunders and G.F. Udell, 1995, The effects of bank capital requirements on bank offbalance sheet financial innovations, *Journal of Banking and Finance*, 19, 647-658.

Jesswein, K., W.R. Folks Jr. and C.C.Y. Kwok, 1996, A framework for analyzing foreign exchange innovations, *Journal of Multinational Financial Management*, 6, 2/3, 31-50.

Jeuland, A., 1981, Parsimonious models for diffusion of innovation: derivation and comparisons, *Working Paper*, Marketing Department, Graduate School of Business, University of Chicago.

Ma, C.K, R.P. Rao and R.L. Peterson, 1989, The resiliency of the high-yield bond market: the LTV default, *Journal of Finance*, 44, 4, 1085-1097.

Mahajan, V. and R.A. Peterson, 1978, Innovation diffusion in a dynamic potential adopter population, *Management Science*, 24, 1589-1597.

Mahajan, V., E. Muller, and F.M. Bass, 1990, New product diffusion models in marketing: a review and directions for future research, *Journal of Marketing*, 54, 1, 1-26.

Mahmood, T., 2000, Survival of newly founded businesses: a log-logistic model approach, *Small Business Economics*, 14, 3, 223-237.

Mansfield, E., 1961, Technical change and the rate of imitation, *Econometrica*, 29, 4, 741-766.

Marqués, D., J. Santillan and A. Rixtel, 2000, Main changes in the financial structure of the euro-zone since the introduction of the euro, *OECD Financial Markets Trends*, 76, 109-127.

Merton, R., 1995, Financial innovation and the management and regulation of financial institutions, *Journal of Banking and Finance*, 19, 461-481.

Miville, M. and A. Bernier, 1999, The corporate bond market in Canada, *Bank of Canada Review*, autumn, 3-8.

Molyneux, P., 1990, US high yield debt and the case for a European market, *National Westminster Bank Quarterly Review*, 3-15.

Molyneux, P. and N. Shanroukh, 1996, Diffusion of financial innovations: the case of junk bonds and note issuance facilities, *Journal of Money, Credit and Banking*, 28, 3, 502-522.

Parker, P.M., 1994, Aggregate diffusion forecasting models in marketing: a critical review, *International Journal of Forecasting*, 10, 2, 353-380.

Rajan, R.G. and L. Zingales, 1998, Financial dependence and growth, *American Economic Review*, 88, 3, 559-586.

Ratkowsky, D., 1990, Handbook of Nonlinear Regression Models, Marcel Dekker.

Srinivasan, V. and C.H. Mason, 1986, Nonlinear least squares estimation of new product diffusion models, *Marketing Science*, 5, 2, 169-178.

Stulz, R.M., 2000, Does financial structure matter for economic growth?, *World Bank Conference on Financial Structure and Economic Development*, Washington D.C. February 10-11, 2000.

Taggart, R.A. Jr., 1988, The growth of the "junk" bond market and its role in financing takeovers, in A.J. Auerbach (ed.), *Mergers and Acquisitions*, NBER Project Report Series, University of Chicago Press, Chicago and London, 5-24.

Tufano, P., 1989, Financial innovation and first-mover advantages, *Journal of Financial Economics*, 25, 213-240.

Winsor, C.P., 1932, A comparison of certain symmetrical growth curves, *Journal of the Washington Academy of Sciences*, 22, 73-84.

Yago, G. and S. Trimbath, 2003, *Beyond Junk Bonds: Expanding High Yields Markets*, Oxford University Press.

Zhang, Z., 2002, Corporate bond spreads and the business cycle, *Bank of Canada Working Paper*, 2002-15.



### European Central Bank working paper series

For a complete list of Working Papers published by the ECB, please visit the ECB's website (http://www.ecb.int).

- 202 "Aggregate loans to the euro area private sector" by A. Calza, M. Manrique and J. Sousa, January 2003.
- 203 "Myopic loss aversion, disappointment aversion and the equity premium puzzle" by D. Fielding and L. Stracca, January 2003.
- 204 "Asymmetric dynamics in the correlations of global equity and bond returns" by L. Cappiello, R.F. Engle and K. Sheppard, January 2003.
- 205 "Real exchange rate in an inter-temporal n-country-model with incomplete markets" by B. Mercereau, January 2003.
- 206 "Empirical estimates of reaction functions for the euro area" by D. Gerdesmeier and B. Roffia, January 2003.
- 207 "A comprehensive model on the euro overnight rate" by F. R. Würtz, January 2003.
- 208 "Do demographic changes affect risk premiums? Evidence from international data" by A. Ang and A. Maddaloni, January 2003.
- 209 "A framework for collateral risk control determination" by D. Cossin, Z. Huang, D. Aunon-Nerin and F. González, January 2003.
- 210 "Anticipated Ramsey reforms and the uniform taxation principle: the role of international financial markets" by S. Schmitt-Grohé and M. Uribe, January 2003.
- 211 "Self-control and savings" by P. Michel and J.P. Vidal, January 2003.
- 212 "Modelling the implied probability of stock market movements" by E. Glatzer and M. Scheicher, January 2003.
- 213 "Aggregation and euro area Phillips curves" by S. Fabiani and J. Morgan, February 2003.
- 214 "On the selection of forecasting models" by A. Inoue and L. Kilian, February 2003.
- 215 "Budget institutions and fiscal performance in Central and Eastern European countries" by H. Gleich, February 2003.
- 216 "The admission of accession countries to an enlarged monetary union: a tentative assessment" by M. Ca'Zorzi and R. A. De Santis, February 2003.
- 217 "The role of product market regulations in the process of structural change" by J. Messina, March 2003.

- 218 "The zero-interest-rate bound and the role of the exchange rate for monetary policy in Japan" by G. Coenen and V. Wieland, March 2003.
- 219 "Extra-euro area manufacturing import prices and exchange rate pass-through" by B. Anderton, March 2003.
- 220 "The allocation of competencies in an international union: a positive analysis" by M. Ruta, April 2003.
- 221 "Estimating risk premia in money market rates" by A. Durré, S. Evjen and R. Pilegaard, April 2003.
- 222 "Inflation dynamics and subjective expectations in the United States" by K. Adam and M. Padula, April 2003.
- 223 "Optimal monetary policy with imperfect common knowledge" by K. Adam, April 2003.
- 224 "The rise of the yen vis-à-vis the ("synthetic") euro: is it supported by economic fundamentals?" by C. Osbat, R. Rüffer and B. Schnatz, April 2003.
- 225 "Productivity and the ("synthetic") euro-dollar exchange rate" by C. Osbat, F. Vijselaar and B. Schnatz, April 2003.
- 226 "The central banker as a risk manager: quantifying and forecasting inflation risks" by L. Kilian and S. Manganelli, April 2003.
- 227 "Monetary policy in a low pass-through environment" by T. Monacelli, April 2003.
- 228 "Monetary policy shocks a nonfundamental look at the data" by M. Klaeffing, May 2003.
- 229 "How does the ECB target inflation?" by P. Surico, May 2003.
- 230 "The euro area financial system: structure, integration and policy initiatives" by P. Hartmann, A. Maddaloni and S. Manganelli, May 2003.
- 231 "Price stability and monetary policy effectiveness when nominal interest rates are bounded at zero" by G. Coenen, A. Orphanides and V. Wieland, May 2003.
- 232 "Describing the Fed's conduct with Taylor rules: is interest rate smoothing important?" by E. Castelnuovo, May 2003.
- 233 "The natural real rate of interest in the euro area" by N. Giammarioli and N. Valla, May 2003.
- 234 "Unemployment, hysteresis and transition" by M. León-Ledesma and P. McAdam, May 2003.
- 235 "Volatility of interest rates in the euro area: evidence from high frequency data" by N. Cassola and C. Morana, June 2003.

- 236 "Swiss monetary targeting 1974-1996: the role of internal policy analysis" by G. Rich, June 2003.
- 237 "Growth expectations, capital flows and international risk sharing" by O. Castrén, M. Miller and R. Stiegert, June 2003.
- 238 "The impact of monetary union on trade prices" by R. Anderton, R. E. Baldwin and D. Taglioni, June 2003.
- 239 "Temporary shocks and unavoidable transitions to a high-unemployment regime" by W. J. Denhaan, June 2003.
- 240 "Monetary policy transmission in the euro area: any changes after EMU?" by I. Angeloni and M. Ehrmann, July 2003.
- 241 Maintaining price stability under free-floating: a fearless way out of the corner?" by C. Detken and V. Gaspar, July 2003.
- 242 "Public sector efficiency: an international comparison" by A. Afonso, L. Schuknecht and V. Tanzi, July 2003.
- 243 "Pass-through of external shocks to euro area inflation" by E. Hahn, July 2003.
- 244 "How does the ECB allot liquidity in its weekly main refinancing operations? A look at the empirical evidence" by S. Ejerskov, C. Martin Moss and L. Stracca, July 2003.
- 245 "Money and payments: a modern perspective" by C. Holthausen and C. Monnet, July 2003.
- 246 "Public finances and long-term growth in Europe evidence from a panel data analysis" byD. R. de Ávila Torrijos and R. Strauch, July 2003.
- 247 "Forecasting euro area inflation: does aggregating forecasts by HICP component improve forecast accuracy?" by K. Hubrich, August 2003.
- 248 "Exchange rates and fundamentals" by C. Engel and K. D. West, August 2003.
- 249 "Trade advantages and specialisation dynamics in acceding countries" by A. Zaghini, August 2003.
- 250 "Persistence, the transmission mechanism and robust monetary policy" by I. Angeloni, G. Coenen and F. Smets, August 2003.
- 251 "Consumption, habit persistence, imperfect information and the lifetime budget constraint" by A. Willman, August 2003.
- 252 "Interpolation and backdating with a large information set" by E. Angelini, J. Henry and M. Marcellino, August 2003.
- 253 "Bond market inflation expectations and longer-term trends in broad monetary growth and inflation in industrial countries, 1880-2001" by W. G. Dewald, September 2003.

- 254 "Forecasting real GDP: what role for narrow money?" by C. Brand, H.-E. Reimers and F. Seitz, September 2003.
- 255 "Is the demand for euro area M3 stable?" by A. Bruggeman, P. Donati and A. Warne, September 2003.
- 256 "Information acquisition and decision making in committees: a survey" by K. Gerling, H. P. Grüner, A. Kiel and E. Schulte, September 2003.
- 257 "Macroeconomic modelling of monetary policy" by M. Klaeffling, September 2003.
- 258 "Interest rate reaction functions and the Taylor rule in the euro area" by P. Gerlach-Kristen, September 2003.
- 259 "Implicit tax co-ordination under repeated policy interactions" by M. Catenaro and J.-P. Vidal, September 2003.
- 260 "Aggregation-theoretic monetary aggregation over the euro area, when countries are heterogeneous" by W. A. Barnett, September 2003.
- 261 "Why has broad money demand been more stable in the euro area than in other economies? A literature review" by A. Calza and J. Sousa, September 2003.
- 262 "Indeterminacy of rational expectations equilibria in sequential financial markets" by P. Donati, September 2003.
- 263 "Measuring contagion with a Bayesian, time-varying coefficient model" by M. Ciccarelli and A. Rebucci, September 2003.
- 264 "A monthly monetary model with banking intermediation for the euro area" by A. Bruggeman and M. Donnay, September 2003.
- 265 "New Keynesian Phillips Curves: a reassessment using euro area data" by P. McAdam and A. Willman, September 2003.
- 266 "Finance and growth in the EU: new evidence from the liberalisation and harmonisation of the banking industry" by D. Romero de Ávila, September 2003.
- 267 "Comparing economic dynamics in the EU and CEE accession countries" by R. Süppel, September 2003.
- 268 "The output composition puzzle: a difference in the monetary transmission mechanism in the euro area and the US" by I. Angeloni, A. K. Kashyap, B. Mojon and D. Terlizzese, September 2003.
- 269 "Zero lower bound: is it a problem with the euro area?" by G. Coenen, September 2003.
- 270 "Downward nominal wage rigidity and the long-run Phillips curve: simulation-based evidence for the euro area" by G. Coenen, September 2003.
- 271 "Indeterminacy and search theory" by N. Giammarioli, September 2003.

- 272 "Inflation targets and the liquidity trap" by M. Klaeffling and V. López Pérez, September 2003.
- 273 "Definition of price stability, range and point inflation targets: the anchoring of long-term inflation expectations" by E. Castelnuovo, S. Nicoletti-Altimari and D. Rodriguez-Palenzuela, September 2003.
- 274 "Interpreting implied risk neutral densities: the role of risk premia" by P. Hördahl and D. Vestin, September 2003.
- 275 "Identifying the monetary transmission mechanism using structural breaks" by A. Beyer and R. Farmer, September 2003.
- 276 "Short-term estimates of euro area real GDP by means of monthly data" by G. Rünstler, September 2003.
- 277 "On the indeterminacy of determinacy and indeterminacy" by A. Beyer and R. Farmer, September 2003.
- 278 "Relevant economic issues concerning the optimal rate of inflation" by D. R. Palenzuela, G. Camba-Méndez and J. Á. García, September 2003.
- 279 "Designing targeting rules for international monetary policy cooperation" by G. Benigno and P. Benigno, October 2003.
- 280 "Inflation, factor substitution and growth" by R. Klump, October 2003.
- 281 "Identifying fiscal shocks and policy regimes in OECD countries" by G. de Arcangelis and S. Lamartina, October 2003.
- 282 "Optimal dynamic risk sharing when enforcement is a decision variable" by T. V. Koeppl, October 2003.
- 283 "US, Japan and the euro area: comparing business-cycle features" by P. McAdam, November 2003.
- 284 "The credibility of the monetary policy 'free lunch" by J. Yetman, November 2003.
- 285 "Government deficits, wealth effects and the price level in an optimizing model" by B. Annicchiarico, November 2003.
- 286 "Country and sector-specific spillover effects in the euro area, the United States and Japan" by B. Kaltenhaeuser, November 2003.
- 287 "Consumer inflation expectations in Poland" by T. Łyziak, November 2003.
- 288 "Implementing optimal control cointegrated I(1) structural VAR models" by F. V. Monti, November 2003.
- 289 "Monetary and fiscal interactions in open economies" by G. Lombardo and A. Sutherland, November 2003.



- 290 "Inflation persistence and robust monetary policy design" by G. Coenen, November 2003.
- 291 "Measuring the time-inconsitency of US monetary policy" by P. Surico, November 2003.
- 292 "Bank mergers, competition and liquidity" by E. Carletti, P. Hartmann and G. Spagnolo, November 2003.
- 293 "Committees and special interests" by M. Felgenhauer and H. P. Grüner, November 2003.
- 294 "Does the yield spread predict recessions in the euro area?" by F. Moneta, December 2003.
- 295 "Optimal allotment policy in the eurosystem's main refinancing operations?" by C. Ewerhart, N. Cassola, S. Ejerskov and N. Valla, December 2003.
- 296 "Monetary policy analysis in a small open economy using bayesian cointegrated structural VARs?" by M. Villani and A. Warne, December 2003.
- 297 "Measurement of contagion in banks' equity prices" by R. Gropp and G. Moerman, December 2003.
- 298 "The lender of last resort: a 21st century approach" by X. Freixas, B. M. Parigi and J.-C. Rochet, December 2003.
- 299 "Import prices and pricing-to-market effects in the euro area" by T. Warmedinger, January 2004.
- 300 "Developing statistical indicators of the integration of the euro area banking system" by M. Manna, January 2004.
- 301 "Inflation and relative price asymmetry" by A. Rátfai, January 2004.
- 302 "Deposit insurance, moral hazard and market monitoring" by R. Gropp and J. Vesala, February 2004.
- 303 "Fiscal policy events and interest rate swap spreads: evidence from the EU" by A. Afonso and R. Strauch, February 2004.
- 304 "Equilibrium unemployment, job flows and inflation dynamics" by A. Trigari, February 2004.
- **305** "A structural common factor approach to core inflation estimation and forecasting" by C. Morana, February **2004**.
- 306 "A markup model of inflation for the euro area" by C. Bowdler and E. S. Jansen, February 2004.
- 307 "Budgetary forecasts in Europe the track record of stability and convergence programmes" by R. Strauch, M. Hallerberg and J. von Hagen, February 2004.
- 308 "International risk-sharing and the transmission of productivity shocks" by G. Corsetti, L. Dedola and S. Leduc, February 2004.
- 309 "Monetary policy shocks in the euro area and global liquidity spillovers" by J. Sousa and A. Zaghini, February 2004.
- 310 "International equity flows and returns: A quantitative equilibrium approach" by R. Albuquerque,G. H. Bauer and M. Schneider, February 2004.
- 311 "Currend account dynamics in OECD and EU acceding countries an intertemporal approach" by M. Bussière, M. Fratzscher and G. Müller, February 2004.

- 312 "Similarities and convergence in G-7 cycles" by F. Canova, M. Ciccarelli and E. Ortega, February 2004.
- 313 "The high-yield segment of the corporate bond market: a diffusion modelling approach for the United States, the United Kingdom and the euro area" by G. de Bondt and D. Marqués, February 2004.

